Author's response to reviews

Title: Boussignac continuous positive airway pressure for the management of acute cardiogenic pulmonary edema: prospective study with a retrospective control group

Authors:

Willem Dieperink (w.dieperink@chir.umcg.nl)
Tiny Jaarsma (t.jaarsma@ thorax.umcg.nl)
Iwan CC van der Horst (iwanouk@gmail.com)
Wybe Nieuwland (w.nieuwland@ thorax.umcg.nl)
Karin M Vermeulen (k.m.vermeulen@mta.umcg.nl)
Hanka Rosman (h.rosman@ thorax.umcg.nl)
Leon PHJ Aarts (l.p.h.j.aarts@ anest.umcg.nl)
Felix Zijlstra (f.zijlstra@ thorax.umcg.nl)
Maarten WN Nijsten (m.w.n.nijsten@chir.umcg.nl)

Version: 5 Date: 18 December 2007

Author's response to reviews: see over
Dear Editor,

Hereby we send our reformatted and accepted manuscript “Boussignac continuous positive airway pressure for the management of acute cardiogenic edema: prospective study with a retrospective control group” with the requested changes.

We detect some errors in figure 1 therefore we hereby send you a new version of this figure.

Yours sincerely,
also on behalf of all authors

Willem Dieperink,